Author's response to reviews

Title: Effect of exercise therapy on lipid profile and oxidative stress indicators in patients with type 2 diabetes

Authors:

Lorenzo Gordon (lorenzogordon@yahoo.com)
Errol Y Morrison (errol.morrison@uwimona.edu.jm)
Donovan A McGrowder (dmcgrowd@yahoo.com)
Ronald Young (ronald.young@uwimona.edu.jm)
Yeiny P Fraser (yeinita@yahoo.com)
Eslaen M Zamoraz (emortorellz@yahoo.es)
Ruby L Alexander-Lindo (lisa.lindo@uwimona.edu.jm)
Rachael Irving (rachael.irving@uwimona.edu.jm)

Version: 7 Date: 12 May 2008

Author's response to reviews: see over
May 7, 2008

Emma Parkin  
Associate Editor,  
BMC-series journals  
Tel: +44 (0)20 7631 9921  
Facsimile: +44 (0)20 7631 9923  
e-mail: editorial@biomedcentral.com  
Web: http://www.biomedcentral.com

Dear Madam,

We have made the necessary changes as suggested in your last e-mail message on May 7, 2008. The list of authors in the manuscript was written exactly as they are in the submission section. The authors were also written identically and in the order in which they are listed in the submission system. Each e-mail address was written on a new line and preceded by the author’s initials.

The ‘Figure legends’ was titled appropriately.

Thank you for your assistance.

Yours sincerely,

Donovan McGrowder, MSc, PhD, FIBMS  
Lecturer in Chemical Pathology  
Department of Pathology  
University of The West Indies